Author's response to reviews

Title: Interventions Aimed at Increasing Research Use in Nursing: A Systematic Review

Authors:

David S Thompson (dst3@ualberta.ca)
Carole A Estabrooks (carole.estabrooks@ualberta.ca)
Shannon Scott-Findlay (shannon.scott-findlay@ualberta.ca)
Katherine Moore (katherine.moore@ualberta.ca)
Lars Wallin (lwa@du.se)

Version: 2 Date: 21 August 2006

Author's response to reviews:

August 18, 2006

Dr Martin Eccles
Editor-in-Chief
Professor of Clinical Effectiveness
University of Newcastle upon Tyne
Centre for Health Services Research
Newcastle upon Tyne
United Kingdom

Dear Professor Eccles,

Please note, I have made changes to an authors affiliation (Dr Lars Wallin) and an authors credentials (Shannon Scott-Findlay)

Sincerely,
David S Thompson, BScN, RN

Knowledge Utilization Studies Program
Masters Student, Faculty of Nursing
5-112 Clinical Sciences Building,
University of Alberta,
Edmonton, Alberta, Canada
T6G 2G3 (780) 492-6836
e-mail: dst3@ualberta.ca